Author's response to reviews

Title: Shared decision making for patients with type 2 patients: a randomized trial in primary care

Authors:

Megan E Branda (branda.megan@mayo.edu)
Annie LeBlanc (leblanc.annie@mayo.edu)
Nilay D Shah (shah.nilay@mayo.edu)
Kristina Tiedje (tiedje.kristina@mayo.edu)
Kari Ruud (ruud.kari@mayo.edu)
Holly Van Houten (vanhouten.holly@mayo.edu)
Laurie Pencille (pencille.laurie@mayo.edu)
Marge Kurland (mkurland@olmmed.org)
Barbara Yawn (byawn@olmmed.org)
Victori M Montori (montori.victor@mayo.edu)

Version: 2 Date: 7 January 2013

Author's response to reviews:

Per the request of the editorial office the following items have been added: The name of the ethics committee is added, under the methods section a statement regarding informed consent is added and the age/sex information in the appendix is removed.